Author’s response to reviews

Title: Coagulopathy as initial manifestation of concomitant celiac disease and cystic fibrosis

Authors:

Aco Kostovski (acokos@gmail.com)
Nikolina Zdraveska (nikolina_389@yahoo.com)

Version: 3 Date: 13 October 2010

Author’s response to reviews:

COVER LETTER FOR SUBMISSION OF MANUSCRIPT

Dear Editor,

We are enclosing a manuscript titled “Coagulopathy as initial manifestation of concomitant celiac disease and cystic fibrosis”
Authors: Aco Kostovski, Nikolina Zdraveska
for publication in “Journal of medical case reports”

We have decided to submit this manuscript because this case report describes a very unusual haemorrhagic presentation of concomitant celiac disease and cystic fibrosis in a 5 year old child.

The chances of these two diseases occurring together is 1: 2 000 000 in the general population. Also, there is no previous report of associated celiac disease and cystic fibrosis initially presented with haemorrhage due to coagulopathy.

Revision was made because the abstract was not structured as specified (introduction, case presentation, conclusion). In the revised manuscript version it corrected.

Thank you for offering the chance of resubmit manuscript.

Sincerely,
Dr.Nikolina Zdraveska